Author's response to reviews

Title: Insulin gene polymorphisms in type 1 diabetes, Addison's disease and the polyglandular autoimmune syndrome type II

Authors:

Elizabeth Ramos-Lopez (elizabeth.ramos-lopez@kgu.de)
Britta Lange (brilange@gmx.net)
Heinrich Kahles (heinrich@ballare.de)
Holger Willenberg (holger.willenberg@uni-duesseldorf.de)
Gesine Meyer (gesine.meyer@kgu.de)
Marissa Penna-Martinez (pennamar@compuserve.de)
Nicole Reisch (Nicole.Reisch@med.uni-muenchen.de)
Stefanie Hahner (hahner_s@medizin.uni-wuerzburg.de)
Jürgen Seissler (jochen.seissler@med.uni-muenchen.de)
Klaus Badenhoop (badenhoop@em.uni-frankfurt.de)

Version: 3 Date: 27 June 2008

Author's response to reviews:

We thank you for the confirmation of acceptance of the manuscript.
All requested changes were made as follow:

Keywords and Short title were removed
Introduction was renamed as Background
Website addresses were removed from the text and cited from the main list
Abbreviations are presented as a list
Reference list contain all authors’ names and the term et al. is not used

Yours sincerely

Klaus Badenhoop M.D, PhD